Author’s response to reviews

Title: Gender differentials in readiness and use of mHealth services in a rural area of Bangladesh

Authors:

Fatema Khatun (kfatema@icddrb.org)
Anita Heywood (a.heywood@unsw.edu.au)
SMA Hanifi (hanifi@icddrb.org)
Shafiqur Rahman (shafiq@isrt.ac.bd)
Predeep Ray (p.ray@unsw.edu.au)
Siaw-Teng Liaw (siaw@unsw.edu.au)
Abbas Bhuiya (abbas.bhuiya@gmail.com)

Version: 2 Date: 14 Jul 2017

Author’s response to reviews:

July 14, 2017

To,
Editor
BMC Health Services Research

Subject: Re: BHSR-D-17-00232R1.Gender differentials in readiness and use of mHealth services in a rural area of Bangladesh

Dear Editor,

Thank you for your concern and providing us opportunity to improve the manuscript. We agree with your concern that several paragraphs in the methodology section of the manuscript were
similar with our other paper published in Plos ONE and IJMI. We have now rewrite those paragraphs (please see page #6 – page #10, page #13 and page #23 in track change version). We believe that revised version addressed the concern regarding similarity index. Please let us know any other concern regarding similarity index according to publication guidelines.

Thank you for your supportive and constructive comments.

Kind regards,

Fatema Khatun
Associate Scientist
Universal Health Coverage
Health Systems and Population Studies Division
icddr,b
68, Shaheed Tajuddin Ahmed Sharani
Mohakhali, Dhaka 1212, Bangladesh
Mobile: +8801715287117
Email: kfatema@icddrb.org
Web: www.icddrb.org